Author's response to reviews

Title: Carcinoid tumor of the verumontanum (colliculus seminalis) of the prostatic urethra with a coexisting prostatic adenocarcinoma: a case report and review of the literature

Authors:

Maxwell Smith MD (Maxwell.Smith@UCDenver.edu)
M Scott Lucia MD (Scott.Lucia@UCDenver.edu)
Francisco G La Rosa MD (Francisco.LaRosa@UCDenver.edu)

Version: 2 Date: 22 September 2009

Author's response to reviews:

Dear JMCR Editors:

We have included de recommended changes in our manuscript:

1. The misspelling of the word "adenocarcinoma" in the Discussion section (after reference [5-6] has been corrected.

2. Added a new figure of the prostatic adenocarcinoma at lower power. The Figure 3 legend has been modified accordingly

Thank you very much for accepting this manuscript in your prestigious journal.

Sincerely,

Francisco G. La Rosa, MD
Pathologist (AP/CP)
Associate Professor
University of Colorado Denver
Department of Pathology
Mail Stop 8104; P.O. Box 6511
12800 East 19th Avenue, Room P18-5124
Aurora, CO 80045-0508

Phone: 303-724-3782
Cell: 303-549-4450